
\noindent Cited entries: \vspace{1em} \\
\noindent

\citet{10.1257/aer.20170619},
\citet{Takeda2019a},
\citet{stakeda2019web},
\citet{saito17:_effec_immig_japan},
\citet{Ryza15:_advan_analy_spark_patter_learn_data_scale},
\citet{takeda2015a},
\citet{Takeda2014a},
\citet{yamazaki13:_japan},
\citet{Takeda2012a},
\citet{bergemann11:_ration},
\citet{goldin:katz:2011},
\citet{takeda10:_cge_analy_welfar_effec_trade},
\citet{10.1257/pol.2.1.155},
\citet{10.1257/aer.100.3.763},
\citet{takeda10:_region_effec_trade_liber_japan},
\citet{2009yamasue502165},
\citet{goldin:katz:2008},
\citet{Biker-2007-unemployment},
\citet{takeda06:_cge_analy_welfar_effec_trade},
\citet{2007yamasue482353},
\citet{Peri2007},
\citet{kusuoka06:_advan_mathem_econom},
\citet{BabikerRutherford-2005-EconomicEffectsof},
\citet{takeda05:_econom_analy_envir_regul},
\citet{ishikawa03:_green_gas_emiss_contr_open_econom},
\citet{brooke03:_gams},
\citet{DeGorter2002},
\citet{Babiker2000525},
\citet{goldin:katz:2000},
\citet{rutherford00:_gtapin_gtap_eg},
\citet{fujita99jp:_spatial_econom},
\citet{Babiker-1999-KyotoProtocoland},
\citet{Babiker-1999-JapaneseNuclearPower},
\citet{Parry1997},
\citet{borgers95:_note_implem_stron_domin},
\citet{wong95:_inter_trade_goods_factor_mobil_},
\citet{ishikawa94:_revis_stolp_samuel_rybcz_theor_produc_exter},
\citet{brezis93:_leapf_inter_compet},
\citet{RePEc:nbr:nberwo:4269},
\citet{krugman91:_geogr_trade},
\citet{helpman91:_inter_trade_trade_polic},
\citet{krugman91:_is_bilat_bad},
\citet{wang89:_model_therm_hydrod_aspec_molten},
\citet{lucas76:_econom_polic_evaluat},
\citet{milne-thomson68:_theor_hydrod},
\citet{essd-10-405-2018},
\citet{luthi08:_high},
\citet{doi:10.1175/2009BAMS2778.1},
\citet{Rivers-2005-CombiningTop-Downand},
\citet{WilsonMannOtsuki-2005-AssessingBenefitsof},
\citet{zhang2016Deep},
\citet{imbens2019Optimized},
\citet{Attwood06:SexedUp_art},
\citet{Attwood09:Mainstreaming},
\citet{attwood2010porn},
\citet{jones84:_handb_inter_econom},
\citet{jones85:_handb_inter_econom},
\citet{jones97:_handb_inter_econom}.
\citet{d'Aspremont-2004-BalancedBayesianMechanisms},
\citet{d'Aspremont-2003-CorrelationIndependenceAndBayesian},
\citet{Chung-1999-ANoteOnMatsushima's},
\citet{d'Aspremont-1998-LinearInequalityMethodsTo},
\citet{myerson98:_exten_poiss_games_condor_jury_theor},
\citet{Black1958},
\citet{Asakawa-2021-DoubleDividendCarbonTax}
\\

\vspace*{1em}
\noindent Articles with certified random order:\\
\noindent
\citet{10.1257/aer.20161492},
\citet{RePEc:hka:wpaper:2018-037},
\citet{vohra18:_maxim_farsig_stabl_set},
\citet{NBERw25205}.

% --------------------
% Local Variables:
% fill-column: 80
% coding: utf-8
% End:

